Author's response to reviews

Title: Caries risk assessment in school children using a reduced Cariogram model without saliva tests

Authors:

Gunnel Hänsel Petersson (gunnel.hansel-petersson@mah.se)
Per-Erik Isberg (per-erik.isberg@stat.lu.se)
Svante Twetman (stwe@sund.ku.dk)

Version: 5 Date: 11 April 2010

Author's response to reviews:

Cover letter April 11, 2010

The following formatting changes have been made to the manuscript:

All tables (Table 1, 2 and 3) are now created using a Tables tool, they are divided into rows and columns orientated in a portrait page layout. All tables are included in the main manuscript. We have also changed the references to the tables throughout the manuscript text.

Figure 1 is on a separate sheet.

We have also checked our manuscript for typographical errors and done final corrections.

Payment has been arranged by sending a scanned, signed Invoice by mail to creditcard@springer.com as we failed to pay by the online system.

With the amendments made, we hope the manuscript is all set for publication in BMC Oral Health.

Yours sincerely,
Gunnel Hänsel Petersson
Corresponding author